## Semi-coupling of a Field-scale Resolving Land-surface Model and WRF-LES to Investigate the Influence of Land-surface Heterogeneity on Cloud Development

Jason Scot Simon<sup>1,1</sup>, Andrew D. Bragg<sup>1,1</sup>, Paul A Dirmeyer<sup>2,2</sup>, and Nathaniel W. Chaney<sup>1,1</sup>

<sup>1</sup>Duke University <sup>2</sup>George Mason University

November 30, 2022

#### Abstract

Contemporary Earth system models mostly ignore the sub-grid scale (SGS) heterogeneous coupling between the land surface and atmosphere, to a detriment that remains largely unknown. To both evaluate the effect of SGS heterogeneity for realistic scenarios and aid in the development of coupled land and atmosphere SGS parameterizations for global models, we present a study of the effect of sub-100 km scale land-surface heterogeneity on cloud development. In the primary experiment we use the Weather Research and Forecasting (WRF) model to conduct two large-eddy simulations over the Southern Great Plains (SGP) site using 100-m horizontal resolution on a domain that spans 100 km in each lateral direction. The first simulation uses high-resolution land-surface fields specified by an offline land-surface model (LSM), while the second uses homogenized land-surface fields found by taking a domain-averaged value of each field at each timestep. The atmospheric development of the heterogeneous and homogeneous simulations are compared, primarily in terms of cloud production and turbulent kinetic energy. It is seen that the heterogeneous case develops a mesoscale circulation pattern which generates additional clouds and turbulence compared to the homogeneous case. Additional experiments isolate sources of heterogeneity in the LSM (including forcing meteorology) to better understand relevant land-surface processes, and modify the Bowen ratio and initial wind profile of the heterogeneous case to clarify the results seen. Finally two additional days at the SGP site are simulated confirming the increase in cloud production in heterogeneous cases.

### Semi-coupling of a Field-scale Resolving Land-surface Model and WRF-LES to Investigate the Influence of Land-surface Heterogeneity on Cloud Development

# Jason S. Simon<sup>1</sup>, Andrew D. Bragg<sup>1</sup>, Paul A. Dirmeyer<sup>2</sup>, and Nathaniel W. Chaney<sup>1</sup>

<sup>1</sup>Duke University <sup>2</sup>George Mason University

#### Key Points:

1

2

3

4

5

6

7

8

1

11

9 •	Large-eddy simulation is used to study fine-scale heterogeneity in land-atmosphere
ð	coupling

- Spatial patterns from rainfall increase cloud production via mesoscale circulations
- Sub-grid scale heterogeneity should ideally be included in global model parameterizations

Corresponding author: Jason Simon, jason.simon@duke.edu

#### 14 Abstract

Contemporary Earth system models mostly ignore the sub-grid scale (SGS) heteroge-15 neous coupling between the land surface and atmosphere. To aid in the development of 16 coupled land and atmosphere SGS parameterizations for global models, we present a study 17 of different aspects of highly-realistic sub-100 km scale land-surface heterogeneity. The 18 primary experiment is a set of simulations of September 24, 2017 over the Southern Great 19 Plains (SGP) site using the Weather Research and Forecasting (WRF) model with 100-20 m horizontal resolution. The overall impact of land-surface heterogeneity is evaluated 21 by comparing cloud and turbulent kinetic energy (TKE) production in large-eddy sim-22 ulations (LESs) using heterogeneous and homogeneous surface fields (namely sensible 23 and latent heat fluxes) specified by an offline field-scale resolving land-surface model (LSM). 24 The heterogeneous land surface leads to significantly more cloud and TKE production. 25 We then isolate specific sources of heterogeneity by using selectively domain-wide aver-26 aged fields in the LSM. It is found that heterogeneity in the land surface created by pre-27 cipitation is effectively responsible for the increases in cloud and TKE production, while 28 rivers and soil type have a negligible impact and land cover has only a small impact. Ad-29 ditional experiments modify the Bowen ratio in the surface fields and the initial wind 30 profile of the heterogeneous case to clarify the results seen. Finally two additional days 31 at the SGP site are simulated showing a similar increase in cloud production in hetero-32 geneous cases. 33

#### <sup>34</sup> Plain Language Summary

A modern Earth system model combines an atmospheric model and land-surface 35 model, and the two interact during a simulation. Due to computational constraints, global 36 models today use grids where very large areas (sometimes in excess of 10,000 square kilo-37 meters) are represented by a single point, making it impossible to directly represent many 38 important features, particularly those related to the development of clouds and rain. Ap-39 proximations of these processes that cannot be represented are included by simpler sub-40 models called parameterizations, which often base calculations on average values over 41 the area they are modeling. To aid in the improvement of these parameterizations, a high-42 resolution model (where each point represents only 0.01 square kilometers) is used to sim-43 ulate three summer days in Oklahoma over a total area of 10,000 square kilometers. It 44 is seen that simulations where the land surface has moist and dry patches from previ-45 ous rain events produce more clouds than simulations where the same amount of soil mois-46 ture is evenly distributed over the entire surface. We hope that this and future work will 47 both motivate and aid efforts to add considerations for the spatial distribution of fea-48 tures, in addition to their average, to the parameterizations used in Earth system mod-49 els. 50

#### 51 **1** Introduction

A critical challenge in characterizing land-atmosphere interactions across scales 52 in Earth system models (ESMs) is the non-linearity that emerges as a result of spa-53 tial heterogeneities over land (Albertson et al., 2001; Bou-Zeid et al., 2004; Huang & 54 Margulis, 2013; Shao et al., 2013; Clark et al., 2015). These complex interactions be-55 tween the land-surface processes and the underlying physical environment drive the 56 spatial complexity of surface fluxes and states (Western et al., 1999; Gómez-Plaza 57 et al., 2001; Jacobs et al., 2004; Chaney et al., 2015). As a result, the macroscale 58 behavior of the water and energy cycles cannot be disentangled from their fine-scale 59 processes and interactions. The heterogeneities that emerge over land, in turn, can 60 play a key role in many important atmospheric processes, such as setting the atmo-61 spheric boundary layer (ABL) depth, initiating convection, and spawning mesoscale 62 circulations (Kustas & Albertson, 2003; Ntelekos et al., 2008; Timmermans et al., 63

2008; Kang & Bryan, 2011; Bertoldi et al., 2013; Gutowski et al., 2020). Further, 64 Weaver (2004b) argue that these effects are non-negligible compared to larger-scale 65 signals over as long as monthly timescales. Although progress is being made in un-66 derstanding the role of multi-scale land heterogeneity on microscale and mesoscale 67 meteorological processes in regional and local studies (Kustas & Albertson, 2003; 68 Talbot et al., 2012; Bertoldi et al., 2013; Huang & Margulis, 2013; Shrestha et al., 69 2014; Senatore et al., 2015), its role in land-atmosphere interactions in the climate 70 system as a whole remains mostly unknown. This is primarily due to the over-71 simplistic coupling between existing sub-grid parameterizations in land-surface 72 models (LSMs) and atmospheric models (e.g., tiling schemes, Ducharne et al., 2000; 73 Bonan et al., 2002; Milly et al., 2014; Chanev et al., 2018; Lawrence et al., 2019). 74 Existing ESMs only exchange sub-grid spatial mean fluxes of mass and energy be-75 tween the land and atmosphere while disregarding higher order sub-grid spatial 76 statistics (e.g., spatial variance). Convection and turbulence parameterizations in at-77 mospheric circulation models are moving towards the inclusion of higher-order SGS 78 processes (e.g., Cloud Lavers Unified By Binormals (CLUBB) and Eddy Diffusivity 79 Mass Flux (EDMF), Golaz et al., 2002; Sušelj et al., 2013), providing an opportunity 80 for potential coupling with the SGS heterogeneity of the land surface. 81

There have been many modeling studies on heterogeneous land surfaces and 82 their effects on atmospheric dynamics, primarily using idealized surface flux fields 83 and initial atmospheres. Pielke Sr (2001) gives a very thorough theoretical back-84 ground and review of the earlier work studying the effect of heterogeneous spatial 85 distributions of sensible and latent heat fluxes from the land surface on the develop-86 ment of cumulus convective rainfall in the atmosphere. The general consensus from 87 LES studies of heterogeneous land-atmosphere interactions is that surface patterns 88 comprised of marked areas of either high sensible heat flux or high latent heat flux 89 (typically resulting from, or an idealization of, underlying patterns of soil moisture 90 and/or vegetation) will lead to secondary mesoscale circulations. These circulations 91 tend to transport moist air from areas with high latent heat fluxes to areas with 92 high sensible heat fluxes where it can be lifted through the ABL, leading to cloud 93 production over the drier land surfaces (Hadfield et al., 1991; Shen & Leclerc, 1995; 94 Avissar & Liu, 1996; Esau & Lyons, 2002; Cheng & Cotton, 2004; van Heerwaarden 95 & de Arellano, 2008; Hohenegger et al., 2009; Garcia-Carreras et al., 2011; Huang & 96 Margulis, 2013; Han, Brdar, Raasch, & Kollet, 2019; Lee et al., 2019). The clouds 97 produced by the aforementioned circulation process tend to be deeper and more lo-98 calized than those produced by homogeneous surfaces, leading to larger overall liquid 99 water path (LWP) values but lower overall cloud cover percentages. 100

The necessary conditions of the land-surface heterogeneity to trigger secondary 101 circulations are not fully established, though it is generally agreed that larger dif-102 ferences between the sensible heat fluxes in the warm and cool patches will produce 103 stronger circulations. It is also generally agreed that the spatial scale of the coher-104 ent warm and cool patches must be of a sufficient size before circulations can be 105 triggered, though with minimal consensus on more specific criteria (Hadfield et al., 106 1992; Chen & Avissar, 1994; Shen & Leclerc, 1995; Albertson et al., 2001; Trier et 107 al., 2004; Patton et al., 2005; Timmermans et al., 2008; Huang & Margulis, 2013; 108 Sühring et al., 2014; Kang, 2016; Kang & Ryu, 2016; Han, Brdar, Raasch, & Kollet, 109 2019; Kang, 2020). Many studies conclude simply that larger spatial scales produce 110 stronger circulations, while others find that there is an optimal scale of land-surface 111 heterogeneity for cloud production after which further increases have a homogenizing effect. The boundary-layer depth is commonly suggested as an optimal scale, though 113 it is argued by van Heerwaarden et al. (2014) that this is too simplistic of a criteria. 114

115 It has been commonly reported by LES studies with idealized surface pat-116 terns that even a modest background wind will effectively eliminate the influence of

land-surface heterogeneity on the atmosphere, implying that scenarios where land-117 surface heterogeneity would notably influence the atmosphere are, in reality, quite 118 limited (Hadfield et al., 1992; Chen & Avissar, 1994; Doran et al., 1995; Avissar & 119 Schmidt, 1998; Eder et al., 2015; Lee et al., 2019). However, idealized LES examples 120 where circulations are maintained with a background wind of 7.5 m s<sup>-1</sup> are shown 121 by Raasch and Harbusch (2001), who explain that claims of wind eliminating cir-122 culations are due to experiments where the surface pattern and wind direction are 123 such that all air parcels are continuously advected over alternating warm and cool 124 patches. Many subsequent studies using idealized surface patterns have confirmed 125 that the orientation of the wind direction compared to the surface heterogeneity 126 pattern determines whether secondary circulations will be eliminated or converted to 127 a rolling structure (Kim et al., 2002; Letzel & Raasch, 2003; Courault et al., 2007; 128 Kang & Lenschow, 2014; Sühring et al., 2014; Rochetin et al., 2017), and multiple 129 studies using surface patterns based on observations have also reported that circu-130 lations are not eliminated by a synoptic wind, but instead develop perpendicular to 131 the prevailing wind direction (Weaver & Avissar, 2001; Weaver, 2004a; Prabha et al., 132 2007; Maronga & Raasch, 2013). Maronga and Raasch (2013) go so far as to state 133 that "[t]he often discussed concept of a blending height, above which the influence 134 of the surface heterogeneity vanishes, thus cannot hold, at least under convective 135 conditions and heterogeneity scales larger than [the boundary-layer depth]." Lynn 136 et al. (1998), via two-dimensional deep convection simulations, found that a strong 137 background wind increased cloud production, owing to a positive feedback between 138 clouds rooted in the ABL and clouds rooted in the middle troposphere. 139

The process of heterogeneous surface fields generating mesoscale circulations 140 which result in cloud production over drier surfaces has also been reported by 141 many observational campaigns (Lyons et al., 1993; Lyons, 2002; Garcia-Carreras 142 et al., 2010; Dixon et al., 2013). Taylor et al. (2011) used satellite observations to 143 study the influence of soil moisture on the development of convective rain storms in 144 West Africa, concluding that soil moisture variations at spatial scales  $\sim 10 - 40$  km 145 strongly control storm development in the region. Taylor et al. (2012) studied the 146 feedback mechanisms between soil moisture and convective storms from global obser-147 vations, finding that drier soils are more likely to produce afternoon rainfall events 148 while wetter soils show no preference for rain development. They note that this 149 result is in contrast to many weather and climate models that use convective pa-150 rameterizations, which show a preference for rainfall development over wetter soils; 151 a modeling study by Hohenegger et al. (2009) demonstrates the tendency of convec-152 tive parameterizations to produce clouds over wetter soils while higher-resolution, 153 convection-resolving models produce clouds over drier soils. Phillips and Klein 154 (2014) studied the Southern Great Plains (SGP) site and found that, while large-155 scale forcings tend to dominate, there are some cases where local feedbacks from 156 the surface play a role in the atmosphere, particularly as soil dries after a precipi-157 tation event. Koster et al. (2003) made an argument for soil moisture heterogeneity 158 effecting precipitation by comparing observations to features in global model results 159 which were known (in the model) to be produced by surface heterogeneity. Low-160 level flight observations by Dixon et al. (2013) found that circulations generated by 161 soil moisture heterogeneity were persistent over the range of observed background 162 winds (up to  $\approx 5 \text{ m s}^{-1}$ ). While the study here focuses primarily on cloud produc-163 tion, the impact of land-surface heterogeneity on the overall Earth system is not 164 isolated to convection. For example, Mendes and Prevedello (2020) suggest, based 165 on analysis of satellite observations, that secondary circulations between patches of 166 different vegetation types has a cooling effect on surface temperatures, and Marsham 167 et al. (2008) used aircraft observations and LES to demonstrate the significance of 168 heterogeneity-induced mesoscale circulations on dust transport in the Sahara. 169

To aid in the development of an effective sub-grid coupling between the mod-170 eled land-surface and atmospheric heterogeneity in ESMs, more must be known 171 about the coupling between land-surface heterogeneity and atmospheric dynamics. 172 While it is generally established that heterogeneous land-surfaces can generate sec-173 ondary circulations which alter cloud characteristics and production rates, many key 174 specifics relevant to parameterizations in global models are unclear. To this end, the 175 study presented here uses output from HydroBlocks, a field-scale resolving LSM, to 176 drive the surface of the Weather Research and Forecasting (WRF) model, run in 177 LES mode, over the SGP site using initial profiles and large-scale heat and moisture 178 fluxes based on observations. The result is a study on the coupling between a real-179 istic land surface and the atmospheric boundary layer (ABL) development over a 180 diurnal cycle, with a specific interest in the role of different sources of land-surface 181 heterogeneity on cloud production at scales which are SGS in a global model. A 182 domain area of  $100 \times 100 \text{ km}^2$  is used, which allows domain-wide mean values to 183 be taken as a representation of a grid-scale value in a global model and the effects 184 of land-surface heterogeneity, which would be SGS on a climate-scale grid, to be 185 studied directly via LES. With this study, we aim to help to answer three key ques-186 tions towards the development of global-scale parameterizations which consider SGS 187 heterogeneity. First, are emergent mesoscale circulations between wet and dry areas 188 observed when using realistic fields for the surface fluxes, background wind, and 189 synoptic fluxes? If so, are the resulting effects significant on the macroscale (domain-190 wide) signal for a domain size comparable to a global model grid cell? Finally, what 191 is the relative impact of the different sources of heterogeneity in the LSM (e.g., soil 192 type, rivers and surface water, soil moisture, etc.) on the macroscale signal? 193

The first experiment here is a pair of simulations of September 24, 2017: the 194 first simulation uses the high-resolution HydroBlocks land surface (described in 195 Sect. 2) while the second spatially homogenizes the land surface by using domain-196 averaged values at each grid point (Sec. 3.1). Cases are then considered where only 197 certain land-surface features are represented heterogeneously in the driving Hy-198 droBlocks simulation, generating different scales of surface heterogeneity (Sec. 3.2). 199 Additional heterogeneous cases are also considered which adjust the Bowen ratio at 200 the surface or the initial wind profile (Sec. 3.3). Finally, the primary heterogeneous 201 vs. homogeneous experiment is repeated for simulations of June 10, 2016 and July 202 16, 2017 with a brief analysis (Sec. 3.4). 203

#### <sup>204</sup> 2 Model description

#### 2.1 WRF

205

Atmospheric simulations are conducted using version 3.8.1 of the WRF model 206 (Skamarock et al., 2008) as an LES (WRF-LES). Model settings largely follow those 207 used in the LES ARM Symbiotic Simulation and Observation Workflow (LASSO) 208 campaign (W. Gustafson et al., 2019; W. I. Gustafson et al., 2020), which is a 209 publicly-available dataset of LES cases over the SGP site. The key difference be-210 tween the LASSO simulations and those presented here is the specification of het-211 erogeneous surface conditions. The LASSO simulations use spatially-uniform, time-212 evolving surface fields for sensible heat flux, latent heat flux, and skin temperature 213 (specified directly), as well as a spatially-uniform and constant momentum drag 214 coefficient. Here, heterogeneous cases use two-dimensional, time-evolving surface 215 fields for sensible heat flux, latent heat flux, skin temperature (found via specified 216 emissivity and upward longwave radiation fields), albedo, and momentum drag coef-217 ficient, all obtained from the HydroBlocks LSM described in Sec. 2.2. The sensible 218 and latent heat fluxes, and drag coefficient are used directly by the WRF dynamics, 219 while the skin temperature, emissivity, and albedo are used by the radiation scheme. 220 As in the LASSO simulations, there is no feedback from the atmosphere to the land 221

surface in the LES; the HydroBlocks LSM is run offline and the output surface fields 222 are specified as the bottom boundary in the WRF model. Other notable differences 223 between the WRF settings used here and those used by LASSO are the expansion 224 of the domain to  $100 \times 100 \text{ km}^2$  (where the LASSO domain is  $25 \times 25 \text{ km}^2$ ), the 225 use of the isotropic three-dimensional Smagorinsky-Lilly turbulence closure model 226 (where LASSO uses the isotropic three-dimensional Deardorff model), the specifica-227 tion of two-dimensional latitude and longitude fields (where LASSO considers every 228 grid point to be at the same latitude and longitude), and the inclusion of a Coriolis 229 forcing (where LASSO specifies f = 0). 230

Following the LASSO configuration, simulations use the Thompson graupel mi-231 crophysics scheme and the RRTMG radiation scheme (though surfaces are specified 232 offline by HydroBlocks) with the cumulus and PBL schemes turned off. The hori-233 zontal resolution is  $\Delta_{x,y} = 100$  m with a timestep of 0.5 s. The domain is approxi-234 mately 14.5 km tall with 227 vertical levels and a vertical resolution of  $\Delta_z = 30$  m in 235 the lower 5 km of the column. Periodic boundary conditions are used in both lateral 236 directions and a w-Rayleigh damping layer is applied in the upper 2 km of the col-237 umn. The LES domain uses a flat bottom boundary, though terrain is considered by 238 the offline HydroBlocks simulation for subsurface and surface routing. Initial profiles 239 for potential temperature, water vapor mixing ratio, and lateral velocity components 240 are obtained from the LASSO database and are applied uniformly to the domain. A 241 relatively unique feature of the LASSO WRF simulations is the inclusion of large-242 scale heat and moisture flux profiles that are applied uniformly on every column in 243 the grid at each timestep as an additional contribution to the respective tendency 244 equations, allowing the use of a single non-nested domain while still providing con-245 siderations for large-scale meteorology. Forcing data for these large-scale fluxes are 246 also obtained from the LASSO database. 247

#### 2.2 HydroBlocks

248

HydroBlocks is a field-scale resolving land-surface model (Chaney, Metcalfe, & 249 Wood, 2016) that accounts for the water, energy, and carbon balance to solve land-250 surface processes at high spatial and temporal resolutions. HydroBlocks leverages 251 the repeating patterns that exist over the landscape (i.e., the spatial organization) 252 by clustering areas of assumed similar hydrologic behaviour into hydrologic response 253 units (HRUs). The simulation of these HRUs and their spatial interactions allows the modeling of the water and energy cycles at field scales (30 m) over regional to 255 continental extents (Chaney, Metcalfe, & Wood, 2016; Chaney et al., 2020; Ver-256 gopolan et al., 2020). The core of HydroBlocks is the Noah-MP vertical land surface 257 scheme (Niu et al., 2011). HydroBlocks applies Noah-MP in an HRU framework 258 to explicitly represent the spatial heterogeneity of surface processes down to field 259 scales. At each timestep, the land-surface scheme updates the hydrologic states at 260 each HRU; and the HRUs dynamically interact laterally via subsurface flow. Fur-261 thermore, the fine-scale river network is modeled via a reach-based kinematic wave with a two-way coupling between the HRUs and their corresponding channels. 263

For this study, HydroBlocks is spun up for two years and uses high-resolution 264 (30 m) soil type and land cover maps from the Probabilistic Remapping of SSURGO 265 (POLARIS) (Chaney, Wood, et al., 2016; Chaney et al., 2019) and National Land 266 Cover Database (NLCD) (Homer et al., 2012) datasets, respectively, and one-eighth 267 degree NLDAS-2 meteorology (Cosgrove et al., 2003; Mitchell et al., 2004) with 268 NCEP Stage-IV radar rainfall ( $\sim 4$  km) data (Lin & Mitchell, 2005). The hourly 269 state of the land surface produced by HydroBlocks for the period of interest is then 270 used to specify surface values in the WRF model for: sensible heat flux, latent heat 271 flux, momentum drag coefficient, albedo, emissivity, and upward longwave radiation. 272 Surface skin temperature is then diagnosed from emissivity and upward longwave ra-273

diation. For the homogeneous cases, skin temperature is diagnosed from mean values
of upward longwave radiation and emissivity, rather than a domain-average of skin
temperature directly. For consistency, surface-flux fields are adjusted so that the
domain-wide averages match the time-evolving scalar surface fluxes specified by the

<sup>278</sup> LASSO campaign, which are from the observationally-improved VARANAL dataset.

#### 279 **3 Results**

Simulations are performed on a  $100 \times 100 \text{ km}^2$  domain over the SGP site, 280 centered at 36.6° N, 97.5° W. The domain is largely cultivated cropland and grass-281 land, with a few small urban areas and a tributary of the Arkansas River running 282 primarily west-east through the domain (Fig. 1). Comparisons between cases are 283 made primarily by evaluating the differences in the development of liquid water 284 path (LWP) in time and space. LWP is of key interest because it serves as a proxy 285 for cloud production and has a high relevance to radiation (Sengupta et al., 2003; 286 Khanal et al., 2020). The LWP and TKE fields presented here are time-averaged 287 values over the previous 10 min interval, sampled every 30 s. On the discretized 288 WRF grid, our measure of LWP is found as 289

$$LWP = \sum_{z} \rho_a q_l \Delta_z \quad [kg m^{-2}]$$
<sup>(1)</sup>

where  $\rho_a$  is moist air density,  $q_l$  is liquid water mixing ratio, and z is the vertical direction.

The temporal development of vertically-integrated, mass-coupled TKE is also compared between cases, serving as a metric for general activity in ABL development. For brevity, hereafter "TKE" may be assumed to refer to the verticallyintegrated, mass-coupled form unless otherwise stated. On the discretized WRF mid\_our unstically integrated mass-coupled TKE is found as

grid, our vertically-integrated measure of mass-coupled TKE is found as

$$\text{TKE} = \sum_{z} \rho_a \left[ \frac{1}{2} \left( u'^2 + v'^2 + w'^2 \right) \right] \Delta_z \quad [\text{kg s}^{-2}]$$
(2)

where u, v, and w are the velocity components in the x (west-east), y (south-north),

and z directions, respectively, and a primed variable indicates deviation from the

<sup>299</sup> mean value in the (x, y) plane.



Figure 1. Map of the simulation domain, centered at the SGP site.

#### 3.1 Heterogeneous vs. homogeneous

300

The primary day considered is September 24, 2017. This day was chosen due 301 to the appreciable spatial heterogeneity in the LSM simulations. Following the 302 LASSO setup, simulations are run for 15 hours beginning at 0538 Local Solar Time 303 (LST) (1200 UTC). Over the  $100 \times 100 \text{ km}^2$  domain, for both the heterogeneous 304 and homogeneous simulations, the average sensible heat flux peaks at  $t \approx 1100$  LST 305 with a magnitude of approximately  $215 \text{ W m}^{-2}$ , and the domain-averaged latent 306 heat flux peaks at the same time with a magnitude of approximately 130 W  $m^{-2}$ 307 (Fig. 2a). In the heterogeneous case the standard deviations of the sensible and la-308 tent heat fluxes both peak at  $t \approx 1300$  LST with values of approximately 40 and 309 45 W m<sup>-2</sup>, respectively (Fig. 2b). Both simulations are initialized with the same 310 domain-wide profiles for potential temperature, water vapor mixing ratio, and lat-311 eral velocity components, shown in Fig. 3. The initial profile is stable with a water 312 vapor mixing ratio of  $\mathcal{O}(10 \text{ g kg}^{-1})$  in the lower 4 km and a wind profile which is 313 predominantly south-north with  $v \approx 15 \text{ m s}^{-1}$  in the lower 10 km of the column. 314 The bulk lateral flow is maintained by the periodic boundary conditions, allowing 315 the profile to develop unconstrained. The large-scale heat and moisture fluxes are 316 predominantly positive influxes over the duration of the simulation, with peak values 317  $\approx 6 \times 10^{-5}$  K s<sup>-1</sup> and  $\approx 4 \times 10^{-5}$  g kg<sup>-1</sup> s<sup>-1</sup>, respectively (not shown). 318

Maps of the surface sensible heat flux and latent heat flux used to drive the 319 WRF-LES surface, upscaled to  $\Delta_{x,y} = 100$  m from the HydroBlocks output, are 320 shown in Fig. 4 at t = 1238 LST, corresponding to the peak standard deviations 321 for sensible and latent heat fluxes in the diurnal cycle. This day was chosen for the 322 large moist patch in the east of the domain, which is a result of a rain event that 323 occurred a few days before. Surface fields in the LES are specified from HydroBlocks 324 every hour (UTC) on the hour and are linearly interpolated in time at each timestep 325 in between. The homogeneous case specifies the domain-averaged value of the afore-326 mentioned surface fields at each grid point, calculated at each timestep. It is worth 327 noting that while the spatial patterns of rivers and subsurface flow are removed in 328

the homogeneous case, their contribution to the domain-wide latent heat flux is still included (Barlage et al., 2021, present a study on the importance of resolving river and stream networks).

The heterogeneous and homogeneous simulations show a notable difference 332 in both domain-wide LWP (Fig. 5a) and vertically-integrated, mass-coupled TKE 333 (Fig. 5b) in time. Both cases begin to produce liquid water in the atmosphere at 334  $t \approx 1000$  LST, but the two cases diverge at  $t \approx 1200$  LST. The heterogeneous case 335 continues to produce liquid water more rapidly, reaching a peak over 300 g m<sup>-2</sup> 336 just after t = 1400 LST, while the homogeneous case has a lower rate of produc-337 tion, reaching a peak of less than 130 g m<sup>-2</sup> near 1500 LST. Production of TKE 338 between the two cases shows similar differences, where the two cases diverge again at 339  $t \approx 1100$  LST with the heterogeneous case reaching a much larger peak value than 340 the homogeneous case. 341

To examine differences in spatial liquid water production, a map of each grid 342 point's maximum LWP value throughout the duration of the simulation is shown for 343 the heterogeneous (Fig. 6a) and homogeneous (Fig. 6b) cases. The heterogeneous 344 case shows a very strong pattern of high liquid water production in the western 30 345 km of the domain and low liquid water production in the eastern 70 km of the do-346 main, while the homogeneous case is very evenly distributed throughout the domain. 347 Recalling that this case has a large moist patch in the east of the domain, cloud pro-348 duction for this case demonstrates a preference for areas with a high sensible heat 349 flux at the surface. This is similar to many previous studies where increased cloud 350 production is achieved by emergent circulation patterns which transport moisture 351 from areas of high latent heat flux to areas of high sensible heat flux where it is 352 then lifted (e.g., Hadfield et al., 1991; Shen & Leclerc, 1995; Avissar & Liu, 1996; 353 van Heerwaarden & de Arellano, 2008; Hohenegger et al., 2009; Huang & Margulis, 354 2013; Rieck et al., 2014; Han, Brdar, & Kollet, 2019; Lee et al., 2019). We will see 355 in Sec. 3.3 that the larger local sensible heat fluxes present in the heterogeneous case 356 alone are not sufficient to generate the levels of cloud production seen. 357

Emergent mesoscale circulations in the heterogeneous case are examined first 358 with cross-section profiles of u(x) at t = 1408 LST, approximately corresponding to 359 the time of peak LWP in the domain, at y = 50 km (Fig. 7a) and averaged over the 360 full domain in the y direction (Fig. 7b). The profiles reveal the anticipated general 361 circulation behavior, where flow is primarily westward in the lower 2 km of the do-362 main with a coherent band of eastward flow aloft which reaches a height of  $z \approx 5$  km 363 at  $x \approx 25$  km, gradually descending to  $z \approx 3$  km over the eastern edge of the do-364 main. The rolling structure induced by the synoptic wind (which is predominantly 365 in the +v direction) is seen very clearly in the *u*-averaged cross-section, centered at 366  $x \approx 20$  km and  $z \approx 1$  km. Similar cross-section profiles across x of potential tem-367 perature and cloud mixing ratio are shown for the heterogeneous case in Figs. 7c-f. 368 Cloud production in the heterogeneous case is focused in the west of the domain 369 reaching an average cloud top just below z = 7 km, with some sparser and lower 370 clouds in the east of the domain. 371

Half-hourly cross-sections of *u*-velocity and cloud mixing ratio are shown for 372 the heterogeneous case along y = 50 km from 1138 to 1538 LST in Fig. 8. At 1138 373 LST, which is just before the LWP time series between the two cases diverge, the 374 profile appears relatively homogeneous, with a band of eastward flow aloft through-375 out the full domain width. By 1238 LST the circulation pattern is clearly visible 376 in the flow, and by 1308 LST it is fully formed with the accompanying cloud pro-377 duction in the west of the domain. At 1438 LST the circulation appears to be in 378 the early stages of decline, and the clouds aloft are spreading laterally and dissi-379 pating, and by 1538 LST both the circulation and cloud layer appear fully in their 380 dissipation phase. 381

Cross-sections along x of u-velocity, potential temperature, and cloud mixing 382 ratio are shown for the homogeneous case in (Fig. 9), also at t = 1408 LST. The 383 y = 50 km cross-section of u(x) for the homogeneous case (Fig. 9a) shows many 384 clear upwelling events, but they are distributed across the full width of the domain 385 without developing any coherent circulation pattern, as is expected of an atmosphere 386 with a uniform surface heating. The cloud mixing ratio similarly shows a very uni-387 form cloud pattern, producing very sparse clouds compared to the heterogeneous 388 simulation with a much lower cloud top, below z = 5 km. 389

390 Compared to the homogeneous case, liquid water production in the heterogeneous case appears to benefit from both the moist and dry patches in its surface 301 forcing, despite them not being co-located, via the latent heat flux from the moist 392 patch being transported laterally to drier areas with a higher sensible heat flux 303 which then lifts the moist air past the lifted condensation level resulting in local 394 cloud production. The homogeneous case, which has the same domain-wide total 395 surface latent and sensible heat fluxes, is unable to generate the same cloud produc-396 tion without local areas of higher sensible heat flux to produce similar local updrafts 397 for the moisture that is present in the boundary layer. The following two sections 398 will further investigate the mechanisms driving the behavior of the heterogeneous 399 case seen here. 400



Figure 2. Time series of the surface sensible heat and latent heat fluxes used for the September 24, 2017 simulations: (a) domain mean for heterogeneous and homogeneous cases, (b) standard deviation for the heterogeneous case.



**Figure 3.** Initial profiles used for the September 24, 2017 simulations: (a) potential temperature, (b) water vapor mixing ratio, (c) *u*-velocity, (d) *v*-velocity.



Figure 4. Heterogeneous surface values for the September 24, 2017 simulations at t = 1238 LST, upscaled from HydroBlocks: (a) sensible heat flux, (b) latent heat flux.



Figure 5. Domain-wide mean fields in time from the heterogeneous and homogeneous September 24, 2017 simulations: (a) LWP, (b) vertically integrated, mass-coupled TKE.



Figure 6. Maximum values of LWP at each grid point throughout the duration of the September 24, 2017 simulations using: (a) heterogeneous land surfaces, (b) homogeneous land surfaces.



Figure 7. Profiles from the September 24, 2017 simulation using heterogeneous surfaces at t = 1408 LST of: (a) *u*-velocity along *x* at y = 50 km, (b) *u*-velocity along *x* and domain-averaged in *y*, (c) potential temperature along *x* at y = 50 km, (d) potential temperature along *x* and domain-averaged in *y*, (d) cloud mixing ratio along *x* at y = 50 km, (f) cloud mixing ratio along *x* and domain-averaged in *y*.



Figure 8. Profiles taken every 30 min from t = 1038 to 1538 LST along x at y = 50 km from the September 24, 2017 simulation using heterogeneous surfaces of: (left column) of u-velocity and (right column) cloud mixing ratio.



Figure 9. Profiles from the September 24, 2017 simulation using homogeneous surfaces at t = 1408 LST of: (a) *u*-velocity along *x* at y = 50 km, (b) *u*-velocity along *x* and domain-averaged in *y*, (c) potential temperature along *x* at y = 50 km, (d) potential temperature along *x* and domain-averaged in *y*, (d) cloud mixing ratio along *x* at y = 50 km, (f) cloud mixing ratio along *x* and domain-averaged in *y*.

#### **3.2** Land-surface components

The heterogeneity in the surface fields used in Sec. 3.1 is the result of four 402 primary sources in the HydroBlocks model: river routing and subsurface flow, soil 403 type, land cover, and forcing meteorology. To better understand the role of land-404 surface heterogeneity in atmospheric dynamics we present four additional WRF 405 simulations which use surface maps from HydroBlocks when considering only cer-406 tain sources of heterogeneity. The first simulation (the "R" case) contains surface 407 heterogeneity generated only by rivers and subsurface flow, using surface fields 408 from a HydroBlocks simulation which calculates river routing and subsurface flow 409 as normal but uses homogenized fields for soil type, land cover, and forcing me-410 teorology. The second simulation ("R+S") follows the same methodology but the 411 driving HydroBlocks simulation also uses the heterogeneous soil-type map. The 412 third simulation ("R+S+LC") uses the heterogeneous land cover field in addition 413 to rivers/subsurface flow and soil type. The fourth simulation ("M") isolates sur-414 face heterogeneity generated by the meteorology driving the LSM by homogenizing 415 the other fields. Each case is energetically constrained so that the domain-averaged 416 surface sensible and latent heat fluxes remain unchanged from the base cases, thus 417 only the standard deviations and spatial scales of heterogeneity differ between these 418 four cases and those in Sec. 3.1. The fully heterogeneous case from Sec. 3.1 is equiv-419 alent to an "R+S+LC+M" case and is used here, along with its corresponding fully 420 homogeneous case, as a reference for comparison. 421

Standard deviations of surface sensible heat flux and latent heat flux in time 422 are shown in Fig. 10a, b, respectively. The sensible heat flux standard deviations 423 are, very approximately, evenly spaced between 10 W m<sup>-2</sup> and 40 W m<sup>-2</sup> with the 424 R case peaking at the lowest value (approximately 10 W m<sup>-2</sup>), followed by the R+S 425 and R+S+LC cases. The M case has a peak standard deviation just below the fully 426 heterogeneous case's peak value (approximately 40 W m<sup>-2</sup>). The latent heat flux 427 standard deviations, on the other hand, have two clear groups: the R, R+S, and 428 R+S+LC cases which have peak values from approximately 10 to 20 W m<sup>-2</sup>, and 429 the fully heterogeneous and M cases which are nearly overlapping with a peak value 430 of approximately 45 W m<sup>-2</sup>. 431

Maps of surface sensible heat flux and latent heat flux at t = 1238 LST for 432 the four cases are shown in Fig. 11. The R case has a largely homogeneous sensible 433 heat flux field (Fig. 11a1) and a river network visible in the latent heat flux field 434 (Fig. 11b1) which, despite appearing very heterogeneous, contains only small spatial 435 scales of heterogeneity and spans the entire domain. The R+S case has a small vi-436 sual increase in sensible and latent heat flux heterogeneity compared to the R case 437 (Fig. 11a2, b2, respectively). The R+S+LC case adds considerable visual detail to 438 the sensible heat flux (Fig. 11a3) and latent heat flux (Fig. 11b3) fields compared 439 to the R+S case. The M case is largely homogeneous in both fields (Fig. 11a4, b4) 440 aside from the  $\sim 50$  km moist patch in the east of the domain, confirming that het-441 erogeneous forcing meteorology is responsible for the larger scales of land-surface 442 heterogeneity seen in the fully heterogeneous case. 443

Considering the resulting time series of LWP and TKE for these cases 444 (Fig. 12a, b, respectively), the R and R+S cases are nearly indistinguishable from 445 the fully homogeneous case while the R+S+LC case follows the fully homogeneous 446 case until  $t \approx 1400$  LST but then has a larger peak than the homogeneous case for 447 both LWP and TKE. The liquid water and TKE production are nearly identical be-448 tween the M and the fully heterogeneous case with very slightly larger values in the 449 M case, indicating that the R, S, and LC sources have a (very small) homogenizing 450 effect on the atmospheric response to the M fields despite the fully heterogeneous 451 case having a slightly larger standard deviation (a similar effect was reported by 452 Zhang et al., 2010). The M case produces nearly all of its liquid water in the west-453

ernmost 40 km of the domain while the other three cases are relatively homogeneous
(not shown).

It is clear in this case that heterogeneous meteorology in the LSM is the primary driver of atmospherically-relevant heterogeneity in the land surface. It is also seen that the standard deviation of surface properties alone is insufficient to describe its impact on atmospheric dynamics, as demonstrated by the close agreement in LWP and TKE production between the fully homogeneous, R, and R+S cases despite significant differences in standard deviations.



**Figure 10.** Time series of standard deviations of the surface fluxes used for the September 24, 2017 simulations where the land model includes heterogeneity from only rivers (R), rivers and soil type (R+S), rivers, soil type and land cover (R+S+LC), and only forcing meteorology (M): (a) sensible heat, (b) latent heat. The fully heterogeneous case is also shown for comparison.



Figure 11. (column a) Surface sensible heat flux and (column b) latent heat flux fields for September 24, 2017 simulations at t = 1238 LST with land surfaces which include heterogeneity from: (row 1) only rivers (R), (row 2) rivers and soil type (R+S), (row 3) rivers, soil type and land cover (R+S+LC), and (row 4) only forcing meteorology (M).



**Figure 12.** Domain-wide mean fields in time from the September 24, 2017 simulations where the land model includes heterogeneity from only rivers (R), rivers and soil type (R+S), rivers, soil type and land cover (R+S+LC), and only forcing meteorology (M): (a) LWP, (b) vertically integrated, mass-coupled TKE. The fully heterogeneous and fully homogeneous cases are also shown for comparison.

#### 3.3 Modified Bowen ratio and wind profile cases

462

It is shown in Sec. 3.1 that a circulation pattern forms between the cool/moist 463 and warm/dry areas of the land surface in the heterogeneous case, however it is 464 not clear how necessary the moisture transported towards the warm surface by this 465 circulation is for the observed cloud production. Patches of high surface sensible 466 heat fluxes relative to their surroundings have been seen in observational and mod-467 eling studies to increase local cloud production without the formation of secondary 468 circulations (e.g., Bosman et al., 2019, and references therein). While this effect is 469 often in the context of heterogeneity created by deforestation, it is possible that the 470 local areas of high sensible heat flux in the heterogeneous surfaces here combined 471 with the moisture that already exists over those areas (from the initial profile and 472 large-scale forcing) are sufficient to generate the increased cloud production of the 473 heterogeneous case without the formation of any secondary circulations. Such a 474 process would also be lost in the homogeneous case, where the sensible heat flux at 475 every grid point is set to the domain-wide mean. To evaluate these two possible ex-476 planations, we consider heterogeneous cases with all of the surface latent heat flux at 177 each grid point converted to additional sensible heat flux at the same grid point (the 478 "0% latent heat" case) and 80% of the surface latent heat flux at each grid point 479 converted to additional sensible heat flux at the same grid point (the "20% latent 480 heat" case). The conversion of latent heat to sensible heat inherently reduces the 481 standard deviation of both the sensible and latent heat flux fields, but without re-482 ducing local maxima of surface sensible heat flux. These two cases, compared to the 483 base heterogeneous case, are used to isolate the effect of larger local sensible heat 484 fluxes on moisture from sources other than surface latent heat fluxes (i.e., from the initial profile and large-scale forcing). 486

Additionally, simulations in previous sections have all used the same initial 487 wind profile which has a south-north component  $v \approx 15 \text{ m s}^{-1}$  throughout the col-488 umn. To evaluate the effect of the wind profile, we consider a case with no wind in 489 the initial profile (the "no wind" case) and a case where the wind at each vertical 490 level of the initial profile is re-oriented to be purely west-to-east (the "w-e wind" 491 case). Both modified-wind cases use the unmodified heterogeneous land surface 492 fields. The motivation for these cases is to add context to the results in Sec. 3.1 493 compared to previous studies available in the literature. 494

Time series of LWP and TKE for all four cases are shown in Fig. 13a, b, re-495 spectively. The increase in surface sensible heat flux in the 0% and 20% latent heat cases slightly speeds up the onset of liquid water production and significantly 497 increases the TKE production in the first six hours compared to the fully heteroge-498 neous and homogeneous cases. The 0% and 20% latent heat cases produce more liq-499 uid water and TKE throughout the simulation than the fully homogeneous case, but 500 are surpassed by the fully heterogeneous case after  $t \approx 1300$  LST. The 20% latent 501 heat case produces slightly more liquid water and TKE than the 0% latent heat case 502 throughout the simulation. The 0% latent heat case produces clouds throughout the 503 domain but with the thickest clouds in the east (Fig. 14a), while the and 20% latent 504 heat cases retains the heterogeneous case's general preference for cloud production in 505 506 the west of the domain (Fig. 14b). It is noteworthy that the 0% and 20% latent heat cases both ultimately produce much less TKE than the heterogeneous case despite 507 significant increases to surface sensible heat flux. It is also interesting that in the 0%508 latent heat case any liquid water produced is solely from moisture that exists in the 509 initial profile or that is introduced by the large-scale moisture flux forcing, both of 510 which are applied uniformly in the domain. Still, the 0% latent heat case is able to 511 produce more liquid water than the fully homogeneous case or any case considered 512 in Sec. 3.2. 513

The no wind case produces more liquid water and TKE than the homogeneous 514 case but less than the heterogeneous case, reaching peak values earlier than both 515 base cases. The w-e wind case shows very similar productions of liquid water and 516 TKE to the base heterogeneous case. Maps of maximum LWP at each grid point 517 throughout the simulation for the two modified wind cases are particularly informa-518 tive. The no wind case (Fig. 14c) produces very concentrated individual clouds of a 519 spatial scale  $\mathcal{O}(1 \text{ km})$  which themselves are very densely distributed in space (when 520 considering the entire 15 hours together) over the entire dry portion of the domain, 521 including the relatively small urban areas in the middle of the moist patch. The 522 w-e wind case (Fig. 14d) shows a strong preference for liquid water production in 523 the southern 20 km of the domain, closely resembling the base heterogeneous case's 524 aversion to cloud production over the moist patch but realigned to the w-e wind 525 direction. 526

The result that clouds show a preferential production over warm/dry areas in 527 the presence of land-surface heterogeneity is widely reported (e.g., Avissar & Liu, 528 1996; Esau & Lyons, 2002; van Heerwaarden & de Arellano, 2008; Hohenegger et 529 al., 2009; Taylor et al., 2012; Huang & Margulis, 2013; Kang, 2016; Lee et al., 2019). 530 The persistence of the circulation with a background wind of the magnitude used 531 here is relatively novel, but the general result that circulations orient themselves 532 perpendicular to the background wind direction has been both demonstrated and 533 explained in previous studies (e.g., Shen & Leclerc, 1995; Raasch & Harbusch, 2001; 534 Weaver, 2004a; Prabha et al., 2007; Sühring et al., 2014; Rochetin et al., 2017), 535 though with less of a consensus. Rieck et al. (2014) state on this topic, in review, 536 that while the role of wind is "controversial", "it is expected that too strong back-537 ground winds mask the effects of land surface heterogeneities". In reality the role of 538 wind is likely even more nuanced than can be effectively evaluated by modern LES 539 studies, e.g., a nested mesoscale-modeling study by Findell and Eltahir (2003) found 540 that the influence of the background wind on surface fluxes, as it relates to trigger-541 ing convection, can either suppress or enhance convection depending on whether it 542 was backing or veering. 543



**Figure 13.** Domain-wide mean fields in time from the 0% LH, 20% LH, no wind, and w-e wind simulations of September 24, 2017: (a) LWP, (b) vertically integrated, mass-coupled TKE. The fully heterogeneous and fully homogeneous cases are also shown for comparison.



Figure 14. Maximum values of LWP at each grid point throughout the duration of modified September 24, 2017 simulations: (a) 0% latent heat case, (b) 20% latent heat case, (c) no wind case, (d) w-e wind case.

#### <sup>544</sup> 3.4 Additional days

To justify the generality of the results seen here for September 24, 2017, two additional days at the SGP site are presented briefly for basic heterogeneous and homogeneous cases. All model settings for these cases are the same as before except for different surface fields, initial soundings, and large-scale forcings. Analysis for these cases is limited to time series of LWP.

Time series of domain-averaged surface sensible heat and latent heat fluxes 550 for the two days are shown in Fig. 15, and maps of the surface sensible and latent 551 heat fluxes at  $t \approx 1230$  LST for the two days are shown in Fig. 16. Both days have 552 land surfaces which are dominated by rainfall from previous days, but in different 553 patterns from each other and from September 24, 2017. The June 10, 2016 case has 554 a surface pattern where moist patches are present in the north-east and south-west 555 corners of the domain (Fig. 16a1, b1), while the moist patch in the July 16, 2017 case dominates the eastern half of the domain (Fig. 16a2, b2). The initial wind pro-557 file for the June 10, 2016 case is similar in magnitude to the September 24, 2017 558 case ( $\approx 15 \text{ m s}^{-1}$ ), while the July 16, 2017 case has only a nominal background wind 559  $(\approx 1 \text{ m s}^{-1})$  (not shown). 560

The time series of domain-wide LWP for heterogeneous and homogeneous simulations of both days (Fig. 17) show similar behavior to the September 24, 2017 simulations, where the heterogeneous cases produce significantly more overall liquid water than their homogeneous counterparts. For all three days, initial liquid

water production is very similar between heterogeneous and homogeneous cases, as 565 is the timing of the ultimate dissipation of liquid water, but large differences are 566 seen midday. The June 10, 2016 case is very similar to the September 24, 2017 case, 567 where the heterogeneous and homogeneous cases begin liquid water production at 568  $t \approx 1000$  LST with very similar rates and then diverge at  $t \approx 1300$  LST when the 569 heterogeneous case accelerates its production and ultimately reaches a much larger 570 peak LWP value than the homogeneous case. The July 16, 2017 case shows a much 571 different behavior, where both the heterogeneous and homogeneous cases show a 572 huge burst of liquid water production at  $t \approx 0800$  LST, with both cases reaching a 573 very similar peak LWP value for the day before 1000 LST. The difference between 574 the two cases for July 16, 2017 is that the homogeneous case shows a rapid decline 575 following the initial burst of liquid water, while the heterogeneous case is able to 576 maintain its liquid water for another four hours. 577

The two additional dates were selected relatively arbitrarily from the LASSO 578 database, which has 95 days from 2015 to 2019 that are pre-selected for shallow 579 convection, based on a spatially and temporally coarse knowledge of standard devi-580 ation in the surface heat fluxes which produced a few dozen candidate dates. That 581 is, they were not chosen with any prior knowledge of how their heterogeneous and 582 homogeneous cases compared. That both of these cases show a similar response in 583 the domain-wide LWP signal to land-surface heterogeneity created by prior rain events, despite ultimately showing very different spatial patterns in the land surface 585 over the same domain, gives a strong indication of the importance of spatial pat-586 terns of heterogeneity in land-atmosphere coupling as it specifically relates to SGS 587 parameterizations for global ESMs. 588



Figure 15. Time series of the domain mean surface sensible heat and latent heat fluxes used for simulations of: (a) June 10, 2016 and (b) July 16, 2017.



Figure 16. (column a) Surface sensible heat flux and (column b) latent heat flux fields at  $t \approx 1230$  LST for simulations of: (row 1) June 10, 2016 and (row 2) July 16, 2017.



Figure 17. Domain-wide LWP in time from the heterogeneous and homogeneous simulations of: (a) June 10, 2016 and (b) July 16, 2017.

#### 589 4 Discussion

Based on multiple decades of both modeling and observational studies, it seems 590 generally apparent that the secondary circulations that can be generated by coherent 591 land-surface patterns, and thus the underlying spatial scale of land-surface hetero-592 geneity, are important for ABL development (Hadfield et al., 1992; Chen & Avissar, 1994; Shen & Leclerc, 1995; Albertson et al., 2001; Letzel & Raasch, 2003; Trier 594 et al., 2004; Patton et al., 2005; Timmermans et al., 2008; van Heerwaarden et al., 595 2014; Sühring et al., 2014; Kang & Ryu, 2016). From the results seen in Sec. 3.1 we 596 can see more specifically the relevance to modern global model dynamics and param-597 eterizations, where heterogeneous land-atmosphere interactions which would be on 598 the SGS in a typical global ESM have a significant impact on the LWP and TKE 599 signals which would be on the grid scale in the same typical ESM. The results in 600 Sec. 3.2 further elucidate the situation, as well as the associated difficulties, by show-601 ing that the most significant driver of SGS land-surface heterogeneity for a global 602 model is SGS atmospheric heterogeneity. This cycle of rainfall patterns which lead 603 to land-surface heterogeneity which triggers deep convection and restarts the cycle 604 has also been suggested in previous studies (Emori, 1998; Lynn et al., 1998; Weaver, 605 2004a, 2004b; Taylor et al., 2012). 606

The LES experiments presented here are an initial investigation into the ef-607 fects of realistic land-surface and atmospheric heterogeneity which is on the SGS 608 in a typical global model, and are intended to be built upon with the ultimate goal 609 of providing useful numerical data for climate-scale diagnostic and parameteriza-610 tion development. The land-surface fields used to drive the LESs are from a diurnal 611 cycle in a spun-up and fully functional LSM using real datasets for land cover, soil 612 type, surface-routing terrain, and meteorology. The fields from the LSM are also 613 assimilated with the observationally-improved VARANAL dataset, further ensur-614 ing realistic energetics in the land surface. The spatial resolution and domain size 615 are both also significant, with  $\Delta_{x,y} = 100$  m over the  $100 \times 100$  km<sup>2</sup> domain and 616  $\Delta_z = 30$  m in the lower 5 km of the vertical column. In this regard, the simulations 617 conducted here offer a significant and novel increase in realism towards the study of 618 the coupling between land and atmosphere heterogeneity in an ESM. However, there 619 are still many idealizations made in the simulations presented which warrant men-620 tioning and examining further in future studies. The two most notable idealizations 621 used here are: the semi-coupled LSM, where the land surface fields are specified a622 priori and do not receive feedback from the atmosphere as it is simulated, and, the 623 periodic lateral boundary conditions. 624

The lack of a feedback between the atmosphere and the land surface means 625 that clouds that develop do not impact the local radiation budget of the land sur-626 face. Rieck et al. (2014), using a fully-coupled LSM, found that the shading effect 627 from clouds reduced the difference between sensible heat fluxes over warm/dry and 628 cool/moist patches by 20%, suggesting that resolved mesoscale circulations would 629 indeed be too strong without atmosphere-to-land coupling. The potential shading 630 feedback may not be as strong in the case considered here, as the case in Rieck et al. 631 (2014) has a minimal background wind speed of 0.5 m s<sup>-1</sup>, compared to the approx-632 imately 15 m s<sup>-1</sup> background wind used here. Such advection of clouds away from 633 their source significantly complicates the nature of atmosphere-to-land feedback (as 634 noted by Weaver & Avissar, 2001). Also, the chessboard pattern used to generate 635 land-surface heterogeneity in Rieck et al. (2014) means that each warm/dry patch is 636 bordered on all four sides by a cool/moist patch, resulting in a much more uniform 637 cloud cover over the warm patches in their study than seen here. Still, the extension 638 of our study to a fully-coupled LSM is a necessary next step which is currently un-639 der development by the co-authors and will provide valuable insights into both the 640 model requirements for an LES with highly heterogeneous land-surface fields and 641

the degree of feedback from atmospheric heterogeneity to the land surface from a physical (though numerical) perspective.

The periodic lateral boundary conditions used in the simulations, while stan-644 dard practice for LES and cloud-resolving studies, is another concession which po-645 tentially influences the results seen here. The cloud production in the no wind case 646 in Sec. 3.3 shows a significant temporal and spatial response to land-surface het-647 erogeneity with a very visible preference for production over drier areas of the land 648 surface. It can thus be assumed that the observed results in the cases with wind are 649 not reliant on the continual recycling of moist air across the domain. However, it 650 is not clear how dependent the results are on the sustained fetches of high sensible 651 heat adjacent to high latent heat that are created by the periodic boundaries. The 652  $100 \times 100 \text{ km}^2$  domain used is large enough to fully encapsulate the moist patch 653 in the September 24, 2017 case, but, while it is plausible to imagine, it cannot be 654 assumed that similar patterns are repeated over the surrounding landscape. In con-655 junction with the fully-coupled simulations mentioned above, nested simulations are 656 also being developed to investigate the influence of the periodic lateral boundary 657 conditions used here. 658

It should also be mentioned that, while  $\Delta_{x,y} = 100$  m is a very high resolution 659 in the cloud-resolving arena, the horizontal resolution does present another potential 660 source for improvement. Multiple idealized studies of the so-called gray zone as it 661 relates to resolving the ABL in an LES have found  $\Delta_{x,y} = 100$  m to be a sufficient 662 horizontal resolution while  $\Delta_{x,y} = 200$  m begins to show signs of grid-dependent 663 turbulence development (e.g., Beare, 2014; Efstathiou & Beare, 2015; J. S. Simon 664 et al., 2019). By this standard, the resolution used here is within the limits of LES. 665 It is not immediately clear how directly these and other idealized gray zone stud-666 ies, which typically use uniform and constant surface sensible heat fluxes, translate 667 to more realistic surface fluxes. In the simulations here, there is a small burst of 668 resolved TKE at  $t \approx 0.000$  LST in most of the cases (e.g., Fig. 5b), which is a com-669 mon characteristic of an artificially delayed onset of resolved turbulence due to the 670 turbulence closure model. Such an artifact does not necessarily indicate that the 671 resolution is irredeemably coarse, so long as the delay of resolved turbulence is not 672 significant and the overall dynamics are accurately simulated once turbulence is trig-673 gered. While the effect of the resolution seen here does not appear to be excessive, 674 the effects of the horizontal resolution cannot fully be appreciated without compar-675 ison to even finer, as well as coarser, simulations of the same case. Such a study is 676 currently under development and is anticipated to provide novel insights towards 677 understanding land and atmosphere heterogeneity, as well as the gray zone of LES 678 turbulence closure models in general. 679

Planned future work generally falls into one or both of two categories: clarify-680 ing the impact of different aspects of the LES configuration on cases with heteroge-681 neous land surfaces, and, providing value to ongoing efforts towards diagnosing and 682 modeling SGS heterogeneity in ESM parameterizations. On the clarification side, 683 the aforementioned three studies (semi- vs. fully-coupled land surfaces, periodic vs. 684 nested lateral boundary conditions, and an expanded range of horizontal resolutions) 685 686 are the top priorities in the near future. Related to aiding diagnostic and parameterization efforts, the most immediate future work is focused on running heterogeneous 687 and homogeneous simulations for multiple dozen additional days, with various ini-688 tial conditions on the surface and in the atmosphere, at the SGP site aided by the 689 available data from the LASSO campaign. In the longer term, we plan to extend 690 simulations to additional locations around the globe where different forms of surface 691 heterogeneity may be studied, e.g., lakes, mountainous terrain, urban areas. These 692 future studies are not exclusive efforts but will be conducted in conjunction with 693 each other, i.e., model configuration choices will be tested for different days and 694

locations, and knowledge gained regarding model behavior will be applied to the diagnostic and parameterization efforts when useful.

#### <sup>697</sup> 5 Summary and conclusions

Realistic land-surface fields are used to evaluate the role of land-surface hetero-698 geneity on atmospheric dynamics, particularly at the grid-scale of a modern global 699 model, by using high-resolution output from the HydroBlocks LSM to specify spa-700 tially heterogeneous and time-evolving surface conditions for sensible heat flux, la-701 tent heat flux, temperature (via emissivity and upward longwave radiation), albedo, 702 and drag coefficient in the WRF model. High-resolution LES cases are then run in 703 a variety of experiments over a domain centered at the SGP site which is sized to 704 mimic a single grid cell in a global model. The primary experiment (Sec. 3.1) com-705 pares two simulations of the diurnal cycle on September 24, 2017: the first using the 706 aforementioned heterogeneous surface fields and the second using time-evolving but 707 spatially-homogeneous surface fields, which take their uniform value of each field as 708 the domain-average of the field in the heterogeneous case. It is observed that the 709 heterogeneous case produces clouds more actively than the homogeneous case and 710 in a spatial pattern that is correlated to the surface sensible heat flux fields. It is 711 shown that the heterogeneous simulation develops a circulation pattern between 712 moist and dry areas where moist air originating over areas of high surface latent 713 heat flux are transported laterally within the boundary layer to areas of high surface 714 sensible heat flux, and are then lifted upwards through the boundary layer leading to 715 cloud production. 716

Experiments are then presented to elucidate the relative impacts of specific 717 sources of land-surface heterogeneity for this case (Sec. 3.2). It is found that spatial 718 patterns created in the land surface by recent rain events are effectively responsible 719 for the entirety of the differences between the atmospheric response in the hetero-720 geneous and homogeneous cases. Contrarily, spatial patterns introduced in the land 721 surface by rivers and subsurface flow have a negligible impact on domain-wide LWP 722 and TKE production compared to the case with the same total surface sensible and 723 latent heat fluxes using fully homogeneous surface values. Spatial patterns from dif-724 ferent soil types also show a negligible impact. Introducing heterogeneity from land 725 cover does introduce a nontrivial increase to both LWP and TKE production com-726 pared to the homogeneous case, however the effect from land cover becomes trivial 727 once heterogeneity from the forcing meteorology is introduced. 728

Modifications of the fully heterogeneous case, where the Bowen ratio at each 729 grid cell is increased and where the background wind profile is changed, are then 730 presented to add clarity to the results seen (Sec. 3.3). It is seen in the cases where 731 the Bowen ratio is increased that the increase in sensible heat is associated with a 732 decrease in cloud production, confirming that the water vapor transported by the 733 emergent circulation pattern is critical for the high rates of cloud production seen 734 in the heterogeneous case. We also see that re-orienting the prevailing wind in the 735 atmosphere will correspondingly re-orient the cloud-production pattern to remain 736 focused over the dry areas, and that removing the mean wind entirely allows clouds 737 to form everywhere that is associated with a high surface sensible heat flux. 738

The last set of additional experiments is a brief analysis of two other summer days at the SGP site, both also with large, but unique, scales of spatial heterogeneity generated by scattered storms at the site on previous days (Sec. 3.4). Both additional days show a significantly larger domain-wide LWP values in the heterogeneous cases compared to their homogeneous counterparts. While the analysis presented here is largely for a single day, these two additional cases strengthen the conclusion that, in the context of SGS parameterizations for global ESMs, landatmosphere coupling of SGS heterogeneity has a significant impact on grid-scalecloud and turbulence production.

From the results seen here, the three questions posed in Sect. 1 can be an-748 swered with relatively high confidence for the SGP area. First, the mesoscale cir-749 culations between wet and dry patches in idealized land surface which have been 750 reported in many LES studies are indeed still present using highly realistic land 751 surfaces and under highly realistic atmospheric conditions, namely a significant 752 background wind. Second, considering an LES domain sized to mimic a single grid 753 cell in a global ESM, the macroscale (domain-wide) atmospheric response to high-754 resolution land-surface heterogeneity is a significant change to total cloud production 755 which should be included in the grid-scale signal. Third, the heterogeneity created in 756 the soil moisture by meteorological patterns has a significant influence on ABL pro-757 cesses (namely cloud and TKE production), while the heterogeneities created in the 758 land surface by rivers and surface water, soil type, and land cover have a relatively 759 small, but non-zero, impact on ABL development. It follows that SGS cloud and 760 turbulence parameterizations for weather and climate models should also include in-761 formation about SGS land-surface heterogeneity and vice versa, though an effective 762 mechanism to do so is yet undeveloped. We hope that this and future work will aid 763 in the development of such mechanisms. 764

#### 765 Acknowledgments

Funded by NOAA grant NA19OAR4310241. PD's contribution is funded by NOAA
 grant NA19OAR4310242.

Data availability: Simulations here use a modification of WRF version 3.8.1 developed and maintained by the LASSO team. The base WRF code, initial sounding files, and large-scale forcing files are available from W. Gustafson et al. (2019).

Additional modifications to the WRF code to specify heterogeneous surfaces, data

files for surface fields for each simulation, and model control files for each simulation

are available at J. Simon and Chaney (2021).

#### 774 **References**

- Albertson, J. D., Kustas, W. P., & Scanlon, T. M. (2001). Large-eddy simulation
   over heterogeneous terrain with remotely sensed land surface conditions. *Water Resources Research*, 37(7), 1939–1953.
- Avissar, R., & Liu, Y. (1996). Three-dimensional numerical study of shallow convective clouds and precipitation induced by land surface forcing. Journal of Geophysical Research: Atmospheres, 101 (D3), 7499–7518.
- Avissar, R., & Schmidt, T. (1998). An Evaluation of the Scale at which Ground-Surface Heat Flux Patchiness Affects the Convective Boundary Layer Using Large-Eddy Simulations. *Journal of the Atmospheric Sciences*, 55(16), 2666-2689. doi: 10.1175/1520-0469(1998)055(2666:AEOTSA)2.0.CO;2
- Barlage, M., Chen, F., Rasmussen, R., Zhang, Z., & Miguez-Macho, G. (2021). The importance of scale-dependent groundwater processes in land-atmosphere interactions over the central united states. *Geophysical Research Letters*, 48(5), e2020GL092171.
- Beare, R. J. (2014). A length scale defining partially-resolved boundary-layer turbulence simulations. *Boundary-Layer Meteorol.*, 151(1), 39–55.
- Bertoldi, G., Kustas, W. P., & Albertson, J. D. (2013). Evaluating source area contributions from aircraft flux measurements over heterogeneous land using large-eddy simulation. *Boundary-layer meteorology*, 147(2), 261–279.
- Bonan, G. B., Oleson, K. W., Vertenstein, M., Levis, S., Zeng, X., Dai, Y., ... Yang,
- <sup>795</sup> Z.-L. (2002). The land surface climatology of the Community Land Model

796	coupled to the NCAR Community Climate Model. Journal of climate, 15(22),
797	
798	Bosman, P. J., van Heerwaarden, C. C., & Teuling, A. J. (2019). Sensible neating
799	as a potential mechanism for enhanced cloud formation over temperate forest.
800	Quarterity Journal of the Royal Meleorological Society, 145(119), 450–408.
801	Bou-Zeid, E., Meneveau, C., & Parlange, M. B. (2004). Large-eddy simulation of
802	height and effective surface neuropass. Water Research (0(2)
803	The second constraints of the second constraints in the second constraints of the second constr
804	Chaney, N. W., Metcane, F., & Wood, E. F. (2010). Hydroblocks: a held-scale re-
805	solving land surface model for application over continental extents. $Hyperbologi-$
806	Change N W Minsery P Herman I D Neuman T W Princard C W
807	Morgan C I Viman V (2010) POLAPIS soil properties: 20 m prob
808	abilistic maps of soil properties over the contiguous United States Water
809	Resources Research 55(4) 2016–2038
010	Change N W Boundy I K Herrera-Estrada I E & Wood E E (2015) High-
811	resolution modeling of the spatial heterogeneity of soil moisture: Applications
012	in network design Water Resources Research $51(1)$ 619–638
013	Chaney N W Torres-Rojas L. Vergopolan N & Fisher C K (2020) Two-way
014	coupling between the sub-grid land surface and river networks in Earth sys-
816	tem models Geoscientific Model Development Discussions 2020 1-31 doi:
817	10.5194/gmd-2020-291
818	Chaney, N. W., Van Huijgevoort, M. H., Shevliakova, E., Malyshev, S., Milly, P. C.,
819	Gauthier, P. P., & Sulman, B. N. (2018). Harnessing big data to rethink land
820	heterogeneity in Earth system models. Hydrology and Earth System Sciences,
821	22(6), 3311-3330.
822	Chaney, N. W., Wood, E. F., McBratney, A. B., Hempel, J. W., Nauman, T. W.,
823	Brungard, C. W., & Odgers, N. P. (2016). POLARIS: A 30-meter probabilistic
824	soil series map of the contiguous United States. Geoderma, 274, 54–67.
825	Chen, F., & Avissar, R. (1994). Impact of land-surface moisture variability on local
826	shallow convective cumulus and precipitation in large-scale models. Journal of
827	Applied Meteorology and Climatology, 33(12), 1382–1401.
828	Cheng, W. Y., & Cotton, W. R. (2004). Sensitivity of a cloud-resolving simulation
829	of the genesis of a mesoscale convective system to horizontal heterogeneities in
830	soil moisture initialization. Journal of Hydrometeorology, $5(5)$ , $934-958$ .
831	Clark, M. P., Fan, Y., Lawrence, D. M., Adam, J. C., Bolster, D., Gochis, D. J.,
832	others $(2015)$ . Improving the representation of hydrologic processes in Earth
833	System Models. Water Resources Research, 51(8), 5929–5956.
834	Cosgrove, B. A., Lohmann, D., Mitchell, K. E., Houser, P. R., Wood, E. F., Schaake,
835	J. C., others (2003). Real-time and retrospective forcing in the North
836	American Land Data Assimilation System (NLDAS) project. Journal of Geo-
837	physical Research: Atmospheres, 108(D22).
838	Courault, D., Drobinski, P., Brunet, Y., Lacarrere, P., & Talbot, C. (2007). Impact
839	of surface heterogeneity on a buoyancy-driven convective boundary layer in
840	light winds. Boundary-layer meteorology, 124 (3), 383–403.
841	Dixon, N., Parker, D., Taylor, C., Garcia-Carreras, L., Harris, P., Marsham, J.,
842	Woolley, A. (2013). The effect of background wind on mesoscale circula-
843	tions above variable soil moisture in the sahel. Quarterly Journal of the Royal
844	Intereor Totogical Society, 139(073), 1009-1024.
845	boran, J., Snaw, W., & Hubbe, J. (1995). Boundary layer characteristics over areas
846	or mnomogeneous surface fluxes. Journal of Applied Meteorology and Climatol- org $2/(2)$ 550 571
847	$\begin{array}{c} \text{Oyy, } \mathcal{O}_{\{2\}}, \text{OST-OTI.}\\ \text{Ducharma A Kostar R D Suproz M I Stignitz M & Kumar D (2000) \\ \end{array}$
848	catchment-based approach to modeling land surface processes in a general cir
850	culation model: 2. parameter estimation and model demonstration. <i>Journal of</i>
	I I I I I I I I I I I I I I I I I I I

Geophysical Research: Atmospheres, 105(D20), 24823–24838. 851 Eder, F., De Roo, F., Rotenberg, E., Yakir, D., Schmid, H. P., & Mauder, M. 852 Secondary circulations at a solitary forest surrounded by semi-arid (2015).853 shrubland and their impact on eddy-covariance measurements. Agricultural 854 and Forest Meteorology, 211, 115–127. 855 Efstathiou, G., & Beare, R. J. (2015). Quantifying and improving sub-grid diffusion 856 in the boundary-layer grey zone. Q J R Meteorol Soc, 141(693), 3006–3017. 857 Emori, S. (1998). The interaction of cumulus convection with soil moisture distribu-858 tion: An idealized simulation. Journal of Geophysical Research: Atmospheres, 859 103(D8), 8873–8884. 860 Esau, I., & Lyons, T. (2002). Effect of sharp vegetation boundary on the convective 861 atmospheric boundary layer. Agricultural and Forest Meteorology, 114(1-2), 3-862 13.863 Findell, K. L., & Eltahir, E. A. (2003).Atmospheric controls on soil moisture-864 boundary layer interactions: Three-dimensional wind effects. Journal of Geo-865 physical Research: Atmospheres, 108(D8). 866 Garcia-Carreras, L., Parker, D. J., & Marsham, J. H. (2011).What is the mech-867 anism for the modification of convective cloud distributions by land surface-868 induced flows? Journal of the Atmospheric Sciences, 68(3), 619–634. 869 Garcia-Carreras, L., Parker, D. J., Taylor, C. M., Reeves, C. E., & Murphy, J. G. 870 Impact of mesoscale vegetation heterogeneities on the dynamical (2010).871 and thermodynamic properties of the planetary boundary layer. Journal of 872 Geophysical Research: Atmospheres, 115(D3). 873 Golaz, J.-C., Larson, V. E., & Cotton, W. R. (2002).A PDF-based model for 874 boundary layer clouds. Part I: Method and model description. Journal of the 875 atmospheric sciences, 59(24), 3540-3551. 876 Gómez-Plaza, A., Martinez-Mena, M., Albaladejo, J., & Castillo, V. (2001). Factors 877 regulating spatial distribution of soil water content in small semiarid catch-878 ments. Journal of hydrology, 253(1-4), 211–226. 879 Gustafson, W., Vogelmann, A., Cheng, X., Dumas, K., Endo, S., Johnson, K., ... 880 Xiao, H. (2019). Description of the LASSO data bundles product. DOE Atmo-881 spheric Radiation Measurement (ARM) user facility. DOE/SC-ARM-TR-216. 882 doi: 10.2172/1469590 883 Gustafson, W. I., Vogelmann, A. M., Li, Z., Cheng, X., Dumas, K. K., Endo, S., ... 884 The Large-eddy simulation (LES) Atmospheric Radiation Xiao, H. (2020).885 Measurement (ARM) Symbiotic Simulation and Observation (LASSO) activity Bulletin of the American Meteorological for continental shallow convection. 887 Society, 101(4), E462–E479. 888 Gutowski, W. J., Ullrich, P. A., Hall, A., Leung, L. R., OBrien, T. A., Patricola, 889 C. M., ... others (2020). The ongoing need for high-resolution regional climate 890 models: Process understanding and stakeholder information. Bulletin of the 891 American Meteorological Society, 101(5), E664–E683. 892 Hadfield, M., Cotton, W., & Pielke, R. (1991). Large-eddy simulations of thermally 893 forced circulations in the convective boundary layer. Part I: A small-scale 894 circulation with zero wind. Boundary-layer meteorology, 57(1), 79–114. 895 Hadfield, M., Cotton, W., & Pielke, R. (1992).Large-eddy simulations of ther-896 mally forced circulations in the convective boundary layer. Part II: The effect 897 of changes in wavelength and wind speed. Boundary-layer meteorology, 58(4), 898 307 - 327.800 Han, C., Brdar, S., & Kollet, S. (2019). Response of convective boundary layer and 900 shallow cumulus to soil moisture heterogeneity: A large-eddy simulation study. 901 Journal of Advances in Modeling Earth Systems, 11(12), 4305–4322. 902 Han, C., Brdar, S., Raasch, S., & Kollet, S. (2019).Large-eddy simulation of 903 catchment-scale circulation. Quarterly Journal of the Royal Meteorological 904 Society, 145(720), 1218–1233. 905

- Hohenegger, C., Brockhaus, P., Bretherton, C. S., & Schär, C. (2009).The soil 906 moisture-precipitation feedback in simulations with explicit and parameterized 907 convection. Journal of Climate, 22(19), 5003–5020. 908 Homer, C. H., Fry, J. A., & Barnes, C. A. (2012). The national land cover database. 909 US Geological Survey Fact Sheet, 3020(4), 1–4. 910 Huang, H.-Y., & Margulis, S. A. (2013).Impact of soil moisture heterogeneity 911 length scale and gradients on daytime coupled land-cloudy boundary layer 912 interactions. Hydrological Processes, 27(14), 1988–2003. 913 Jacobs, J., Mohanty, B., Hsu, E., & Miller, D. (2004).Field scale variability and 914 similarity of soil moisture. Remote Sens. Environ, 92, 436–446. 915 Kang, S.-L. (2016). Regional Bowen ratio controls on afternoon moist convection: 916 A large eddy simulation study. Journal of Geophysical Research: Atmospheres, 917 121(23), 14,056-14,083. doi: 10.1002/2016JD025567 918 Kang, S.-L. (2020). Effects of mesoscale surface heterogeneity on the afternoon and 919 early evening transition of the atmospheric boundary layer. Boundary-Layer 920 Meteorology, 174(3), 371-391.921 Kang, S.-L., & Bryan, G. H. (2011).A large-eddy simulation study of moist con-922 vection initiation over heterogeneous surface fluxes. Monthly weather review, 923 139(9), 2901-2917.924 Kang, S.-L., & Lenschow, D. H. (2014). Temporal evolution of low-level winds in-925 duced by two-dimensional mesoscale surface heat-flux heterogeneity. Boundary-926 layer meteorology, 151(3), 501-529. 927 Kang, S.-L., & Ryu, J.-H. (2016). Response of moist convection to multi-scale sur-928 face flux heterogeneity. Quarterly Journal of the Royal Meteorological Society, 929 142(698), 2180-2193.930 Khanal, S., Wang, Z., & French, J. R. (2020). Improving middle and high latitude 931 cloud liquid water path measurements from modis. Atmospheric Research, 932 243, 105033. 933 Kim, H.-J., Noh, Y., & Raasch, S. (2002). Interaction between wind and tempera-934 ture fields under the heterogeneous heat flux in the planetary boundary layer. 935 In 15th conference on boundary layer and turbulence. 936 Koster, R. D., Suarez, M. J., Higgins, R. W., & Van den Dool, H. M. (2003). Obser-937 vational evidence that soil moisture variations affect precipitation. *Geophysical* 938 Research Letters, 30(5). 939 Kustas, W. P., & Albertson, J. D. (2003). Effects of surface temperature contrast on 940 land-atmosphere exchange: A case study from monsoon 90. Water Resources 941 Research, 39(6). 942 Lawrence, D. M., Fisher, R. A., Koven, C. D., Oleson, K. W., Swenson, S. C., Bo-943 nan, G., ... others (2019). The Community Land Model version 5: Description 944 of new features, benchmarking, and impact of forcing uncertainty. Journal of 945 Advances in Modeling Earth Systems, 11(12), 4245–4287. 946 Lee, J. M., Zhang, Y., & Klein, S. A. (2019).The effect of land surface hetero-947 geneity and background wind on shallow cumulus clouds and the transition to 948 deeper convection. Journal of the Atmospheric Sciences, 76(2), 401-419. 949 Large eddy simulation of thermally induced Letzel, M. O., & Raasch, S. (2003).950 oscillations in the convective boundary layer. Journal of the atmospheric sci-951 ences, 60(18), 2328–2341. 952 Lin, Y., & Mitchell, K. E. (2005). The NCEP stage II/IV hourly precipitation anal-953 yses: Development and applications. In Proceedings of the 19th conference hy-954 drology, american meteorological society, san diego, ca, usa (Vol. 10). 955 Lynn, B. H., Tao, W.-K., & Wetzel, P. J. (1998). A study of landscape-generated 956 deep moist convection. Monthly weather review, 126(4), 928–942. 957 Lyons, T. (2002). Clouds prefer native vegetation. Meteorology and Atmospheric 958 *Physics*, 80(1), 131–140. 959
- Lyons, T., Schwerdtfeger, P., Hacker, J., Foster, I., Smith, R., & Xinmei, H. (1993).

961	Land-atmosphere interaction in a semiarid region: The bunny fence experi-
962	ment. Bulletin of the American Meteorological Society, 74(7), 1327–1334.
963	Maronga, B., & Raasch, S. (2013). Large-eddy simulations of surface heterogeneity
964	effects on the convective boundary layer during the litfass-2003 experiment.
965	Boundary-Layer Meteorology, $146(1)$ , $17-44$ .
966	Marsham, J. H., Parker, D. J., Grams, C. M., Johnson, B. T., Grey, W. M., & Ross,
967	A. N. (2008). Observations of mesoscale and boundary-layer scale circulations
968	affecting dust transport and uplift over the sahara. Atmospheric Chemistry
969	and Physics, 8(23), 6979–6993.
970	Mendes, C. B., & Prevedello, J. A. (2020). Does habitat fragmentation affect
971	landscape-level temperatures? A global analysis. Landscape Ecology, 35(8),
972	1743–1756.
973	Milly, P. C., Malyshev, S. L., Shevliakova, E., Dunne, K. A., Findell, K. L., Gleeson,
974	T., Swenson, S. (2014). An enhanced model of land water and energy
975	for global hydrologic and earth-system studies. Journal of Hydrometeorology,
976	15(5), 1739–1761.
977	Mitchell, K. E., Lohmann, D., Houser, P. R., Wood, E. F., Schaake, J. C., Robock,
978	A., others (2004). The multi-institution North American Land Data As-
979	similation System (NLDAS): Utilizing multiple GCIP products and partners in
980	a continental distributed hydrological modeling system. Journal of Geophysical
981	Research: Atmospheres, $109(D7)$ .
982	Niu, GY., Yang, ZL., Mitchell, K. E., Chen, F., Ek, M. B., Barlage, M., oth-
983	ers (2011). The community Noah land surface model with multiparameteriza-
984	tion options (Noah-MP): 1. model description and evaluation with local-scale
985	measurements. Journal of Geophysical Research: Atmospheres, 116(D12).
986	Ntelekos, A. A., Smith, J. A., Baeck, M. L., Krajewski, W. F., Miller, A. J., &
987	Goska, R. (2008). Extreme hydrometeorological events and the urban en-
988	vironment: Dissecting the 7 july 2004 thunderstorm over the Baltimore MD
989	Metropolitan Region. Water Resources Research, 44(8).
990	Patton, E. G., Sullivan, P. P., & Moeng, CH. (2005). The influence of idealized
991	heterogeneity on wet and dry planetary boundary layers coupled to the land
992	surface. Journal of Atmospheric Sciences, 62(7), 2078–2097.
993	Phillips, T. J., & Klein, S. A. (2014). Land-atmosphere coupling manifested in
994	warm-season observations on the US southern great plains. Journal of Geo-
995	physical Research: Atmospheres, 119(2), 509–528.
996	Pielke Sr, R. A. (2001). Influence of the spatial distribution of vegetation and soils
997	on the prediction of cumulus convective rainfall. Reviews of Geophysics, $39(2)$ ,
998	151 - 177.
999	Prabha, T. V., Karipot, A., & Binford, M. W. (2007). Characteristics of secondary
1000	circulations over an inhomogeneous surface simulated with large-eddy simula-
1001	tion. Boundary-layer Meteorology, $123(2)$ , $239-261$ .
1002	Raasch, S., & Harbusch, G. (2001). An analysis of secondary circulations and
1003	their effects caused by small-scale surface inhomogeneities using large-eddy
1004	simulation. Boundary-layer meteorology, $101(1)$ , $31-59$ .
1005	Rieck, M., Hohenegger, C., & van Heerwaarden, C. C. (2014). The influence of land
1006	surface heterogeneities on cloud size development. Monthly Weather Review,
1007	142(10),  3830 - 3846.
1008	Rochetin, N., Couvreux, F., & Guichard, F. (2017). Morphology of breeze circula-
1009	tions induced by surface flux heterogeneities and their impact on convection
1010	initiation. Quarterly Journal of the Royal Meteorological Society, 143(702),
1011	463 - 478.
1012	Senatore, A., Mendicino, G., Gochis, D. J., Yu, W., Yates, D. N., & Kunstmann,
1013	H. (2015). Fully coupled atmosphere-hydrology simulations for the central
1014	Mediterranean: Impact of enhanced hydrological parameterization for short
1015	and long time scales. Journal of Advances in Modeling Earth Systems, $7(4)$ ,

1016	1693 - 1715.
1017	Sengupta, M., Clothiaux, E. E., Ackerman, T. P., Kato, S., & Min, Q. (2003).
1018	Importance of accurate liquid water path for estimation of solar radiation in
1019	warm boundary layer clouds: An observational study. Journal of climate.
1020	16(18), 2997–3009.
1021	Shao Y Liu S Schween J H & Crewell S (2013) Large-eddy atmosphere-
1021	land-surface modelling over heterogeneous surfaces: Model development and
1022	comparison with measurements $Boundary-layer meteorology 1/8(2) 333–$
1024	
1025	Shen S & Leclerc M Y (1995) How large must surface inhomogeneities be
1025	before they influence the convective boundary layer structure? a case study
1027	Quarterly Journal of the Royal Meteorological Society 121(526) 1209–1228
1020	Shrestha P Sulis M Mashou M Kollet S & Simmer C (2014) A scale-
1020	consistent terrestrial systems modeling platform based on COSMO CLM and
1029	ParFlow Monthly weather review 1/2(9) 3466–3483
1030	Simon I & Chaney N (2021 May) Data for: Semi-coupling of a Field-
1031	scale Resolving Land-surface Model and WRF-LES to Investigate the Influ-
1032	ence of Land-surface Heterogeneity on Cloud Development Zenodo doi:
1034	10 5281/zenodo 4741327
1034	Simon I S Zhou B Mirocha I D & Chow F K (2010) Evolicit filtering
1035	and reconstruction to reduce grid dependence in convective boundary layer
1030	simulations using WBE-LES Man Weather Rev. $1/7(5)$ 1805-1821 doi:
1029	10 1175/MWB-D-18-0205 1
1030	Skamarock W Klemp I Dudhia I Gill D Barker D Duda M Powers
1040	I (2008) A description of the Advanced Research WRF version 3 NCAR
1040	Technical Note NCAR/TN-175+STR
1041	Sühring M Maronga B Herbort F & Baasch S (2014) On the effect of surface
1042	heat-flux heterogeneities on the mixed-layer-top entrainment Boundary-layer
1043	meteorology 151(3) 531–556
1045	Sušeli K Teiveira I & Chung D (2013) A unified model for moist convective
1045	boundary layers based on a stochastic eddy-diffusivity/mass-flux parameteriza-
1047	tion Journal of the Atmospheric Sciences $70(7)$ 1929–1953
1049	Talbot C Bou-Zeid E & Smith J (2012) Nested mesoscale large-eddy simula-
1040	tions with WRF: Performance in real test cases Journal of Hudrometeorology
1050	13(5), 1421-1441.
1051	Taylor C M de Jeu B A Guichard F Harris P P & Dorigo W A (2012)
1052	Afternoon rain more likely over drier soils Nature 189(7416) 423–426
1053	Taylor C M Gounou A Guichard F Harris P P Ellis B J Couvreux F &
1054	De Kauwe M (2011) Frequency of Sahelian storm initiation enhanced over
1055	mesoscale soil-moisture patterns. <i>Nature Geoscience</i> , $\lambda(7)$ , 430–433.
1056	Timmermans, W., Bertoldi, G., Albertson, J., Olioso, A., Su, Z., & Gieske, A.
1057	(2008). Accounting for atmospheric boundary layer variability on flux estima-
1058	tion from rs observations. International journal of remote sensing, 29(17-18).
1059	5275-5290.
1060	Trier, S. B., Chen, F., & Manning, K. W. (2004). A study of convection initia-
1061	tion in a mesoscale model using high-resolution land surface initial conditions.
1062	Monthly weather review, 132(12), 2954–2976.
1063	van Heerwaarden, C. C., & de Arellano, J. V. G. (2008). Relative humidity as an
1064	indicator for cloud formation over heterogeneous land surfaces. Journal of the
1065	Atmospheric Sciences, $65(10)$ , $3263-3277$ .
1066	van Heerwaarden, C. C., Mellado, J. P., & De Lozar, A. (2014). Scaling laws for
1067	the heterogeneously heated free convective boundary laver. Journal of the At-
1068	mospheric Sciences, 71(11), 3975–4000.
1069	Vergopolan, N., Chaney, N. W., Beck, H. E., Pan, M., Sheffield, J., Chan. S., &
1070	Wood, E. F. (2020). Combining hyper-resolution land surface modeling with

SMAP brightness temperatures to obtain 30-m soil moisture estimates. *Remote Sensing of Environment*, 242, 111740.

Weaver, C. P. (2004a). Coupling between large-scale atmospheric processes and mesoscale land-atmosphere interactions in the us southern great plains during summer. part i: Case studies. *Journal of Hydrometeorology*, 5(6), 1223–1246.

1071

1072

1080

1081

1082

- Weaver, C. P. (2004b). Coupling between large-scale atmospheric processes and mesoscale land-atmosphere interactions in the us southern great plains during summer. Part II: Mean impacts of the mesoscale. *Journal of Hydrometeorology*, 5(6), 1247–1248.
  - Weaver, C. P., & Avissar, R. (2001). Atmospheric disturbances caused by human modification of the landscape. Bulletin of the American Meteorological Society, 82(2), 269–282.
- Western, A. W., Grayson, R. B., Blöschl, G., Willgoose, G. R., & McMahon, T. A.
   (1999). Observed spatial organization of soil moisture and its relation to
   terrain indices. *Water resources research*, 35(3), 797–810.
- Zhang, N., Williams, Q. L., & Liu, H. (2010). Effects of land-surface heterogeneity
   on numerical simulations of mesoscale atmospheric boundary layer processes.
   Theoretical and applied climatology, 102(3), 307–317.